Enhancing the Prioritization of Disease-Causing Genes through Tissue Specific Protein Interaction Networks

Oded Magger¹*, Yedael Y. Waldman¹, Eytan Ruppin^{1,2}, Roded Sharan¹

1 Blavatnik School of Computer Science, Tel Aviv University, Tel Aviv, Israel, 2 School of Medicine, Tel Aviv University, Tel Aviv, Israel

Abstract

The prioritization of candidate disease-causing genes is a fundamental challenge in the post-genomic era. Current state of the art methods exploit a protein-protein interaction (PPI) network for this task. They are based on the observation that genes causing phenotypically-similar diseases tend to lie close to one another in a PPI network. However, to date, these methods have used a static picture of human PPIs, while diseases impact specific tissues in which the PPI networks may be dramatically different. Here, for the first time, we perform a large-scale assessment of the contribution of tissue-specific information to gene prioritization. By integrating tissue-specific gene expression data with PPI information, we construct tissue-specific PPI networks for 60 tissues and investigate their prioritization power. We find that tissue-specific PPI networks considerably improve the prioritization results compared to those obtained using a generic PPI network. Furthermore, they allow predicting novel disease-tissue associations, pointing to sub-clinical tissue effects that may escape early detection.

Citation: Magger O, Waldman YY, Ruppin E, Sharan R (2012) Enhancing the Prioritization of Disease-Causing Genes through Tissue Specific Protein Interaction Networks. PLoS Comput Biol 8(9): e1002690. doi:10.1371/journal.pcbi.1002690

Editor: Donna K. Slonim, Tufts University, United States of America

Received October 15, 2011; Accepted July 28, 2012; Published September 27, 2012

Copyright: © 2012 Magger et al. This is an open-access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Funding: OM and YYW were supported in part by a fellowship from the Edmond J. Safra Bioinformatics program at Tel Aviv University. YYW was also supported by Eshkol Fellowship from the Israeli Ministry of Science and Technology. ER and RS were supported by a Bikura grant from the Israel Science Foundation and a James Mcdonnel Foundation grant. RS was further supported by a research grant from the Israeli Science Foundation (grant no. 241/11). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing Interests: The authors have declared that no competing interests exist.

* E-mail: OdedMagger@gmail.com

Introduction

A fundamental challenge in human health is elucidating the molecular basis of hereditary diseases. Contemporary methods for discovering disease-causing genes usually consist of two steps: first, genome-wide association studies identify genomic intervals that are linked to a disease of interest. Second, the genes within these intervals are examined for their causal relation to the disease [1–3]. Experimentally verifying that a gene is associated with a disease is an expensive and time-consuming process, calling for the prioritization of candidate causal genes. A plethora of computational methods were developed to meet this challenge. These methods are often based on system-wide data such as protein interaction networks [4–9], gene expression [8,10–12], sequence similarity of genes [13,14], functional similarity and annotation [8,12,13] and more (for a review on these methods see [15,16]).

Many state of the art algorithms for the gene prioritization problem use protein interaction or functional linkage networks [8,17,18], exploiting the tendency of genes causing similar diseases to lie close to one another in the network [16,19,20]. However, these methods do not take into account the fact that the vast majority of genetic disorders tend to manifest only in a single or a few tissues [20–22]. Typically, the same data sets are used to prioritize genes for a liver disease or a brain disease, even though the molecular landscapes of a Hepatocyte and a neuron are quite dissimilar.

In this work, we incorporate tissue-specific gene expression data into the prioritization process and demonstrate its impact on the prioritization results. The integration is achieved by constructing tissue-specific protein-protein interaction (PPI) networks and employing them in the prioritization. The rationale behind this approach is that many disorders involve a disruption of the 'molecular fabric' of different, healthy tissues. From a protein interaction network point of view, this disruption can be often characterized as a perturbation of a gene, corresponding to node removal, or the perturbation of an interaction between two gene products, corresponding to an edge removal [23]. In the context of genetic disorders, even though the underlying harmful mutation exists in all the cells of our body, it most often wreaks havoc only in a few tissues. This tissue selectivity is likely to emerge due to differences in the functionality of the mutated protein within these tissues, its tissue-specific interacting proteins, its abundance and the abundance of its interactors. Thus, the hypothesis underlying the current work is that a tissue specific network, which better represents the actual disease-related tissue, is likely to yield more accurate prioritizations of the diseases it manifests.

The concept of tissue-specific protein interactions is relatively unexplored. Bossi and Lehner [24] analyzed human PPIs in a tissue-specific context, showing that many housekeeping proteins interact with highly tissue-specific proteins, which in turn implies that housekeeping proteins may have tissue-specific roles. Emig and Albrecht [25] expanded this analysis to identify functional differences between tissues, showing that tissue-specific protein interactions are often involved in transmembrane transport and receptor activation. Lin et al [26] analyzed the topological properties of housekeeping and tissue specific proteins within the generic (non tissue-specific) PPI network. Waldman et al. [27] analyzed translation efficiency in humans using PPIs. Using tissue specific PPI networks, they showed that proteins whose genes are translated more efficiently in a specific tissue tend to have more

Author Summary

Identifying the genes causing genetic disease is a key challenge in human health, and a crucial step on the road for developing novel diagnostics and treatments. Modern discovery methods involve genome-wide association studies that reveal regions of the genome where the causal gene is likely to reside, and then prioritizing the candidate genes within these regions and experimentally examining the most promising candidates' potential influence on the disease. Many computational methods were developed to automatically prioritize candidate genes. Some of the most successful methods use a biological network of interacting genes or proteins as an input. However, these networks - and subsequently, these methods - do not take into account the differences between tissues. In other words, a heart disease is analyzed using the same network as a skin disease. We constructed tissue-specific protein interaction networks and explored their effect on an existing prioritization algorithm by comparing the algorithm's performance on the tissue-specific networks and the generic network. We find that integrating tissue-specific data indeed leads to better prioritization. We also used the prioritization results of different tissues in order to suggest new disease-tissue associations.

connections within this tissue as compared to other proteins in the same tissue. Lopes et al. [28] created unweighted tissue-specific networks for several separate PPI databases. They used these networks to analyze host-pathogen interactions in a tissue-specific manner. Finally, a proof-of-concept work by Jiang et al. [29] combined five tissue-specific networks taken from Bossi and Lehner [24] to prioritize candidate genes for type 2 Diabetes.

Of note, the lack of tissue specific PPI networks stands in marked difference from the existence of many tissue- and cell-specific variants of other types of biological networks, such as regulatory networks [30–32], functional linkage networks [33,34] and metabolic networks [35–37].

The current study is the first large-scale study that aims to enhance the accuracy of existing network-based gene prioritization algorithms by taking into account tissue-specific information. This is achieved by constructing tissue specific PPI networks and utilizing them for gene prioritization instead of the standard, generic PPI network. First, we examine the hypothesis that a gene is likely to be expressed in a healthy tissue for its mutation to clinically manifest in that tissue. Indeed, a large majority (71-83%) of the known disease-causing genes are significantly expressed in the corresponding disease-associated tissue. However, not all disease-associated genes are significantly expressed in the tissues where the disease is manifested. Interestingly, as shown below, we find that most of the remaining genes either have a low expression level across all tissues, or are involved in mediating a response to external stimulus or being involved in multi-cellular developmental processes, and as such are not expected to have high expression under steady-state conditions in the adult tissue.

Focusing on the cases where the disease-related gene is expressed in the associated tissue, we show that integrating tissue specific expression information into a gene prioritization scheme markedly improves its prediction accuracy. Specifically, we generate tissue-specific PPI networks for 60 healthy human tissues using gene expression data from those tissues [38]. We then apply the same candidate prioritization algorithm for both the original and the tissue-specific PPI networks, and compare the performance of each in a cross-validation setting. We find that the tissuespecific variant of the algorithm yields higher area under the receiver-operator characteristics (ROC) curve (AUC) and gives the correct gene a higher ranking than the original variant more often than not. Finally, we extend our method to infer new disease-tissue associations.

Results

Tissue-specific expression of disease causing genes

We constructed literature-based gene-disease and disease-tissue association sets. To this end, we retrieved a set of known genedisease associations from GeneCards [4,39]. Disease-tissue associations were based on an association matrix generated computationally by Lage et al. [40]. This matrix provides disease-tissue association scores based on the co-occurrence of disease-related and tissue-related terms in PubMed abstracts. These scores are normalized per disease and presented as percentages (See Methods).

For each disease, we assigned the tissue that had the **maximal association score** (**MAS**) with that disease, and filtered diseases whose MAS was below a predefined threshold. For most of the following analyses, we used two thresholds: MAS>8% was the cutoff used by Lage et al., estimated by them to provide 80% assignment accuracy. Filtering by this threshold produced a set of 920 disease-gene associations, spanning 729 diseases and 632 genes. The second threshold, MAS>40%, was estimated by Lage et al to provide \sim 90% accuracy. This threshold yielded 349 associations spanning 290 diseases and 269 genes. In both cases, genes whose related disease could not be associated with a specific tissue were removed from the analysis.

Next, we constructed binary tissue-specific gene expression profiles for 60 healthy tissues based on the Novartis Research Foundation Gene Expression Database (GNF) [38] (Methods). Out of 9998 proteins composing the *generic* (not tissue-specific) network, the number of proteins expressed in each tissue varied between 1322 (13%) to 7113 (71%; mean = 4500.8, standard deviation = 1399.3; Supp. Table S1).

For each gene-disease association, we checked whether the causal gene is expressed in the tissue assigned to the disease. Interestingly, we found that a considerable fraction of the causal genes were not expressed in their assigned tissue, ranging between 29% and 17% from MAS>8% to MAS>60%, respectively (Figure 1). Importantly, this fraction is significantly smaller than that expected by chance (38.25% lowly-expressed genes are expected on average across all MAS thresholds, p<1E-5; Methods).

To better understand why disease-causing genes might be lowly expressed in their associated tissues, we studied in detail the 76 lowly-expressed disease-causing genes under a MAS threshold of 40%. First, we analyzed the functional annotations of those genes. Notably, 44 (58%) of the genes were found to be involved in multicellular development processes (GO:0007275, FDR E-value: 1.8E-11), where 36 of those were directly involved in organ development (GO:0048513, FDR E-value: 7.1E-12). Hence, mutations in these genes might disrupt their early embryonic activity leading to pathologies in adult tissues regardless of their expression in these mature tissues. In addition, 17 (22%) of the genes were involved in cellular response to stimulus (GO: 0051716, FDR E-value: 1.8E-4) and, therefore, may not be expressed under normal conditions.

We also found that disease-causing genes that were lowly expressed in the tissue associated to the disease tended to be expressed in fewer tissues than expected (12.1 tissues on average compared to 17.5 at random, p < 1E-5; Methods). In addition,



Figure 1. The fraction of disease genes expressed in the disease's assigned tissue correlates with MAS Threshold. The fraction of disease-causing genes expressed in the tissue of their pertaining disease, compared to the random expectation (obtained through a permutation test; Methods), for different MAS thresholds. The error bars represent the minimal and maximal fraction of expressed genes observed at random (over 10,000 permutations) for each MAS threshold. Total number of associations is (from lower to higher MAS): 920, 812, 583, 349 and 167.

doi:10.1371/journal.pcbi.1002690.g001

these genes exhibited lower mRNA levels across all tissues than the expected levels (150.4 versus 224.8 Affymetrix average difference (AD) units expected by chance, p < 1E-5, see Methods). We believe that these observations may partly explain the phenomenon of low-expression of genes in the pertaining disease tissues, as further elaborated upon in the Discussion section. Henceforth, we focused on the majority of disease-causing genes where the gene is indeed expressed in its associated tissue (denoted 'the expressed disease genes association set').

Constructing tissue-specific protein-protein interaction networks

We considered two methods for converting the generic PPI network into a tissue-specific network using a given tissue-specific expression profile. These methods are summarized in Figure 2 and discussed below.

A naïve method, titled "Node Removal" (NR), was used previously by Bossi and Lehner [24], Waldman et al [27] and Lopes et al [28]. By this method, a tissue-specific PPI network is generated by removing from the network proteins that are not expressed in the relevant tissue. Notably, such a scheme dramatically changes the connectivity of the network: while a generic PPI networks tends to have a single giant connected component containing most of the network's nodes [41], the NR network is shattered into a relatively small giant component and many small connected components. While the generic human network is composed of 91 connected components with the giant component covering 98% of the network (9796 proteins), the resulting tissue-specific networks have 545 connected components on average, with the average size of the giant component being 3907 proteins. The other components are very small - a few surpass 4 proteins, and none surpass 10.

The number of interactions also drops, from 41049 in the generic network to 14257.21 on average (Range: 2026[4.9%]–27571[67.1%], standard deviation: 6195.4). The amount of expressed proteins and retained interactions in the network have a strong positive correlation (Pearson: p = 0.9939). Moreover, there's also a similarly strong positive correlation between the amount of expressed proteins and average interactions per expressed protein at the tissue (Pearson: p = 0.9803), suggesting that the power-law distribution of interactions is retained. See

Supp. Table S1 for the detailed properties of the tissue-specific Node Removal networks.

The second tissue-specific network reconstruction method, novel to this work, is titled **'Edge Reweight'** (**ERW**). By this method, we do not alter the topology of the generic network, but rather modify the edge weights to reflect the probability that the corresponding interactions take place at the specific tissue. In brief, the original confidence score of an edge is multiplied by a penalty factor, nw, for each interacting protein that is not expressed in the tissue (see Methods for full details). Note that when nw = 0, the ERW network becomes the NR network; conversely, when nw = 1, the ERW network is identical to the original PPI network. Thus, varying values of nw allow us to control just how tissue specific the network is.

The NR and ERW (with rw = 0.1) PPI networks are publicly available as supporting material (Datasets S1, S2, S3).

Predicting causal genes using tissue-specific Protein interaction networks

In order to prioritize candidate disease genes, we used the PRINCE prioritization algorithm, which we have previously shown to compare favorably to other state-of-the-art algorithms [4,17]. For completeness, we include a brief description of PRINCE below; for a detailed description see [4].

PRINCE receives a weighted PPI network, a disease-disease phenotypic similarity network and a disease-gene association set as inputs. Given a query disease, PRINCE assigns a prior score to genes associated with known diseases that are phenotypically similar to the query. This score is then propagated through a PPI network in an iterative process, culminating in a smooth scoring function where the score of a node tends to be similar to the scores of its neighboring nodes.

In more detail, let q be the query disease and denote by F(v) the prioritization score to be computed for gene v. Let Y(v) be the prior score for gene v (with respect to q), defined as $Y(v) = disease_similarity(q,d)$ if v is associated with a disease d, and Y(v) = 0 if no disease is associated with v. If v is known to be associated with multiple diseases, then disease that is most similar to q is chosen.

F(v) is calculated as a linear combination of $\Upsilon(v)$ and the scores of *v*'s neighboring nodes:

$$F(v) = \alpha \left[\sum_{u \in N(v)} F(u) \cdot w(u, v) \right] + (1 - \alpha) \cdot Y(v)$$

Where $\mathcal{N}(v)$ is the set of nodes adjacent to v in the network, w(u,v) is the confidence of the interaction between u and v, and $\alpha \in [0,1]$ is a parameter controlling the relative importance of the network vs. the prior information.

We applied PRINCE to score disease-causing genes using both the original PPI network and the tissue-specific networks built with the **NR** and **ERW** strategies, and used a leave-one-out cross validation to assess the performance of PRINCE given each network as input, in terms of AUC (Methods). For ERW, we used the MAS>40% association set as a benchmark to identify the optimal *rw* parameter, by constructing multiple TS-ERW (Tissue-Specific Edge ReWeight) networks with varying values of *rw* and comparing their AUCs (Figure S1 and Methods). By Figure S1, the performance has a single peak situated at the lower end of *rw's range:* **rw** = 0.1 for the entire disease-gene association set, and **rw** = 0.001 for the expressed disease genes association set. In the following results presentation we will concentrate on these two choices of the *rw* parameter, as well as on rw = 0.5 which is situated



Figure 2. A summary of tissue-specific PPI network reconstruction methods. First, we determine the set of expressed genes in a given tissue based on an expression cutoff of 200 Affymetrix AD units. The set of expressed genes is then superimposed on the general PPI using one of two strategies: (a) Node Removal – removing genes which are considered unexpressed from the network. (b) Edge Reweight - Reducing the weight of an edge connecting one or two unexpressed genes. This results in a tissue specific PPI network. doi:10.1371/journal.pcbi.1002690.g002

in mid-range and thus represents a moderately tissue-specific network. As explained above, we focused the performance evaluation on the subset of disease-gene associations where the causal genes are known to be expressed in the associated tissue (see Figure S3 and Text S1 for an analysis over the entire association set).

For MAS>40%, the AUC of the original, generic PPI network (0.825) was lower than that of each representative tissue-specific network (0.85–0.88). The results, summarized in Figure 3 and Figure S2, point to a moderate yet considerable improvement. Among the tissue-specific networks, **TS-ERW** with rw = 0.5, which is the most similar to the original network, exhibits the smallest improvement. The improvement peaks for **TS-ERW** with rw = 0.001. **TS-NR** (tissue-specific node removal) and **TS-ERW** with rw = 0.1 networks have comparable AUC values.

We further inspected the cross-validation results by comparing the ranking of true causal genes in the generic network to the tissue-specific networks on a case-to-case basis, in order to estimate how often the tissue-specific data improves the prioritization. Instead of bundling all of the cross-validation results together, we regarded every test case (disease-gene association) in the data set separately, and compared the ranking given to the actual causal gene by PRINCE using the generic and the tissue-specific PPI networks. We found that for every tested MAS threshold, both ERW and NR tissue-specific PRINCE ranked true associations higher than the generic PRINCE in a majority of the cases (Table 1). This also holds true when considering the entire association set, with the exception of NR at MAS>8% (Supp.



Figure 3. Comparing generic and tissue-specific PPIs' performance in candidate disease genes prioritization. Performance comparison between the generic and different variants of tissuespecific PRINCE, according to the ROC Area under curve (AUC) of causal gene prediction in a leave-one-out cross validation test. Error bars represent the standard deviation of AUC values obtained when replacing leave-one-out with 25-fold cross validation of ten random partitions. Results are for a disease-tissue MAS threshold of 40%. doi:10.1371/journal.pcbi.1002690.q003

Table S2). For example, when choosing a MAS threshold of 8% (the same threshold used in [40]) and reweight parameter nw = 0.1, we observe that TS-ERW PRINCE gives better ranks to 288 (47%) true associations, whereas the generic network PRINCE gives better ranks to only 58 (9.5%) true associations. 266 associations are identically ranked under both network types. To assess the significance of the improved rankings, we performed a Wilcoxon signed-rank test between the rankings of true causal genes provided by generic PRINCE and every tissue-specific variant. As evident from Table 1, the rankings obtained by the tissue-specific variant (p<1E-8). Similar trends were observed when analyzing the entire association set (Table S2).

Inferring disease-tissue associations

Having the ability to predict the effects of disease genes on specific tissues, naturally gives rise to the question: given a disease (a collection of disease-causing genes), what tissues are most likely to be affected? This is of particular interest, since while the overt clinical manifestations of a disease are usually well-known, in many cases it may have more subtle, sub-clinical tissue effects that may escape early detection. Such alterations may manifest themselves at later stages of the disease, and may be wrongly attributed to other potential complications and confounding factors, instead of the original disease, which can serve at least as an important predisposing factor.

To investigate this potential scenario in depth, we developed a method to computationally infer disease-tissue associations using the framework presented in the previous section. For a given query disease, we applied TS-ERW PRINCE ($\pi v = 0.1$, chosen for its robustly positive results for both disease-gene association sets presented in the paper) once for every tissue, using the tissue's modified PPI network as input. We then ranked the tissues according to the **relative rank** PRINCE assigns to the causal gene in the tissue' respective networks (for results obtained using the **absolute score** PRINCE assigns to the causal gene see Supp. Text S1 and Figure S4). For example, given a disease *d* and a known causal gene *g*, if g is ranked 4th, when using PRINCE with the Kidney PPI and 6th, when using PRINCE with the Heart PPI, then the kidney is considered more strongly associated with *d* than the heart.

We compared our predicted disease-tissue associations to the data collected by Lage et al. For every disease with MAS>40%,



Figure 4. Evaluation of tissue-disease association prediction. The histogram shows the distribution of our disease-tissue ranking for the tissues assigned by Lage et al. in every test case (disease-gene association). As can be seen, in more than half of the cases the associated tissue was predicted first among all other tissues. doi:10.1371/journal.pcbi.1002690.q004

we checked what ranking was given to the tissue which was assigned the highest association score by Lage et al (Figure 4). In 53% of the cases the tissue predicted by Lage was ranked first by us as well (p<0.013, see Methods). These results further show the power of tissue specific PPI approach to detect tissue specific disease involvement. Obviously, such analyses could not have been performed using the generic PRINCE method, which is oblivious of the tissue-specific information.

Discussion

In the current study we aimed to infer disease causing genes using tissue-specific PPI networks. Most previous studies that used these networks to infer causal genes were based on generic PPI networks and ignored differences between tissues [19]. Nevertheless, this generic approach may be limited as there are significant differences in expression patterns between tissues, both with respect to mRNA as well as in protein levels [38,42]. These differences imply that different tissues have different active PPI networks: a specific interaction may take place in some tissues while not in others, based on the expression distribution of the interacting proteins [24]. Moreover, these differences may explain why, in many cases, a disease may affect a specific tissue and not others: the same protein may be active in specific tissues and

		#cases of better ranking			
MAS threshold	Tissue-specific network type	Tissue-specific	Tie	Generic	Wilcoxon signed-rank test p-value
8%	NR	295	203	103	2.09e-15
8%	ERW, <i>rw</i> = 0.001	291	233	88	9.12e-26
8%	ERW, <i>rw</i> = 0.1	288	266	58	1.88e-37
8%	ERW, <i>rw</i> = 0.5	248	334	30	8.85e-37
40%	NR	125	91	40	7.68e-9
40%	ERW, <i>rw</i> = 0.001	124	102	30	1.24e-14
40%	ERW, <i>rw</i> = 0.1	122	117	17	2.84e-17
40%	ERW, <i>rw</i> = 0.5	103	145	8	6.34e-17

Table 1. Evaluation of generic and tissue-specific gene prioritization methods according to their ranking of the true causal genes.

The table presents a case-to-case comparison of the ranking provided by generic and tissue-specific PRINCE, as well as the statistical significance of this comparison using Wilcoxon signed-rank test.

doi:10.1371/journal.pcbi.1002690.t001

inactive in others, or can have different function in different tissues based on its different neighbors in the different networks. Following these observations, we decided to examine the utility of building and incorporating tissue specific PPI networks in our analysis. Adding tissue specificity information for various diseases [40] we were able to perform a tissue specific inference of disease causing genes.

We used the PRINCE algorithm for gene prioritization and contrasted between generic and tissue specific PPIs. We found that the tissue specific approach enhances the performance of the algorithm. In our analysis we used two different methods for tissue specific PPI networks construction that yield different gene prioritization performance. We observed that better results were obtained when modifying the weights of the networks edges (using the ERW method) compared to following the more drastic approach of removing lowly-expressed proteins from the network (using the NR method). There may be several explanations for these differences. First, it may be related to PRINCE algorithm. A global network-based algorithm such as PRINCE is expected to be less successful when applied to a more disconnected network, such as those generated by the NR approach. Moreover, even for other algorithms that are based on local inference which is not propagated, ERW may be proven more appropriate. NR is a very strict method, eliminating every unexpressed protein, while ERW assigns a continuous value for the interaction based on the expression of the two interacting proteins. Thus, the former is likely to be less robust to noisy data such as gene expression [43-45]).

One might suggest that there is no need to generate tissue specific PPI networks for tissue specific prioritization. Rather, one might use the generic PRINCE and then, in a post-processing manner, assign the lowest possible ranks to the lowly-expressed genes in the tissue being investigated. While such an attenuation approach performs poorly when applied to the entire gene-disease association set (AUC = 0.755, Figure S3, 'Unexpressed genes attenuated'), it improves over the standard PRINCE when its application is restricted to associations that involve genes that are expressed in the relevant tissue. As shown in Figure S5, for MAS>40% this approach yields an AUC of 0.897, albeit significantly smaller than the attenuated version of TS-ERW (AUC = 0.905, Wilcoxon sign rank P-value = 0.02 for nw = 0.1).

Interestingly, as a preprocessing step for the tissue specific PRINCE algorithm, we found that a considerable fraction of disease genes are not expressed in the tissue associated with the disease. There may be several explanations for this observation. First, it may reflect an error in measurements, either of the expression microarray or the computational inference of diseasetissue association. Nevertheless, such a substantial fraction of genes is more likely to reflect a true biological observation. For example, a protein may be active although having lower mRNA levels. Posttranscriptional modifications or higher translational efficiency may also result in higher protein levels or longer protein half-lives [46,47]. In addition to putative differences between mRNA and protein levels, obviously, there may be proteins who perform their function in relatively low levels Indeed, we found that many of the genes unexpressed at their disease's assigned tissue also have low overall expression levels, suggesting that these genes might still be expressed at functional level in the diseased tissue.

Another possibility may be that the damage to the tissue was caused by a disruption of the protein function within the tissue in earlier developmental stages. Supporting this hypothesis we found that lowly-expressed disease causing genes are enriched with developmental annotations such as multicellular development processes (GO:0007275) and organ development (GO:0048513), and with stimulus response annotations (GO:0051716). Hence, the protein may not be active in the adult tissues (as manifested by its expression pattern), but a mutation in the genes may alter normal development of the tissue or may prevent the normal response of the tissue to stress or other stimuli, resulting in a disease. Finally, due to the complexity and the dependencies between tissues in a multi-tissue organism, a mutation in a protein active in one tissue may result in clinical pathology in another tissue. For example, Vitamin D – dependent rickets 1A (MIM: 264700) is primarily a bone disorder, but it is caused by a mutation in the gene *CTP27B1*, which is active in the kidney and participates in the hydroxylation of Vitamin D into its active form, Calcitriol [48]. Overall, the role of lowly expressed genes in causing disease in a given tissue is a rather complex one and deserves a separate analysis that is beyond the current scope.

Some limitations of the current analysis should be mentioned. First, a direct tissue specific measure of protein abundance would be more adequate than mRNA levels as a measure for the presence and hence the activation and functionality of a protein in a tissue. However, despite the best efforts of the scientific community, compendiums of human tissue-specific protein abundance levels across multiple tissues are not nearly as comprehensive as the mRNA expression dataset we use, both in tissue scope, gene coverage and quantitative resolution [42]. In addition, the mutual expression of two possibly interacting proteins does not guarantee that the interaction will take place, and there are other factors that also should be considered such as, most prominently, the proteins' phosphorylation state. Nevertheless, even given these limitations, our tissue specific approach performs better than the generic approach. As large scale data on tissue specific protein abundance and cellular localization will become available, it will be interesting to repeat the analysis reported here to see whether it yields better predictions, as may be expected.

In recent years, PPI networks were shown to be a powerful tool in many fields of molecular biology, such as predicting protein annotation and more [19,49,50]. We hope that the results of this study will encourage future studies to utilize tissue-specific PPI networks to further increase their predictive and explanatory power.

Methods

Tissue specific gene expression

We downloaded the Novartis Research Foundation Gene Expression Database (GNF) tissue-specific gene expression data set [38] from the Gene Expression Omnibus (http://www.ncbi. nlm.nih.gov/geo/) (GDS596). We averaged replicas and for each gene took the probe with the maximal expression level. We used 60 non-diseased tissues for which disease-tissue association data existed (out of 79 tissues). Following [24,51], a gene was considered expressed in a tissue if its expression in that tissue was at least 200 Affymetrix average-difference (AD) units. Similarly, a protein was defined to be present in a tissue if the corresponding gene was expressed there.

Disease-tissue associations

The disease-tissue association matrix was contributed by Kasper Lage [40]. Lage et al. estimated the association of a tissue and a disease by measuring their co-occurrences in PubMed abstracts. Specifically, the association score was computed using Ochiai's coefficient (OC) [52], and then normalized by the sum of all OCs for the same disease. Subsequently, each disease was associated with the tissue that attained the highest association score. A computationally-generated disease-tissue association set was used since at this time there was no large-scale, manually curated disease-tissue association set available.

Analysis of lowly expressed causal genes

For 76 disease genes that were lowly-expressed or not expressed (i.e., expression below 200 AD units) in the tissue associated with the relevant disease, we conducted functional enrichment analysis using the DAVID web server [53,54]. To test if these genes tend to be lowly expressed across all tissues, we generated a random set of genes that are lowly-expressed on the same disease tissues as the original set. I.e., for each original disease-causing gene that is lowly expressed in the associated tissue, we randomly selected another gene that is also lowly-expressed in that tissue. Next, we computed for each gene the number of tissues in which it was expressed (expression breadth) and compared the resulting distribution to that of a random set, repeating the comparison across 10,000 random sets built similarly. In the same manner we also evaluated the significance of the distribution of average expression level of the lowly-expressed genes.

To estimate the number of disease genes that are expected to be lowly-expressed at an assigned tissue at random, we computed this quantity for 10,000 permutations of the tissue assignment vector taken for a given MAS threshold. We permuted the vector instead of picking a tissue from a uniform distribution for every disease, in order to maintain the bias of tissues that tend to be associated with many diseases (e.g. skin diseases, cardiac diseases). Since the fraction measured experimentally was lower than those resulting from the 10,000 permutations, the estimated p-value is p < 1E-5.

Generic and tissue-specific network construction

We constructed a weighted human PPI network with 9,998 proteins and 41,702 interactions. The network is based on three high throughput experiments [55–57] and the HPRD database [58]. The interactions were assigned confidence scores based on the experimental evidence available for each interaction using a logistic regression model adapted from [59]. We considered two ways of obtaining tissue-specific networks: **Node-removal tissue-specific PPI network** was derived by removing from the original PPI network proteins that are not expressed in the relevant tissue, and all of the edges adjacent to them. The remaining edges were retained, along with their weights. In an **edge – reweight tissue-specific PPI network**, the confidence of each interaction represents the probability that the interaction takes place within a given tissue.

We now describe in detail the reweighting scheme that we used. Our underlying assumption was that an interaction between proteins P_1 and P_2 occurs at a specific tissue t if only if P_1 and P_2 interact in the general network and are both expressed at tissue t. Denote the event that proteins P_i and P_j interact in the generic network as $I_{i,j}$, and the event that protein P_i is expressed in tissue t as X(i,t). Now, a gene is considered expressed in a given tissue if its measured expression level in that tissue is above 200 AD units. However, expression data is often noisy [45,51] so there is a chance that a gene not passing this cutoff is still expressed (we assume that if a gene passes the threshold then it is indeed expressed in the given tissue). If we denote this probability by *rw*, then

$$\begin{split} & w'_{i,j} \!=\! P\big(P_i, P_j \text{ interact} | \text{Tissue} \!=\! t\big) \!= \\ & P\big(I_{i,j}|t\big) * P(X(i,t)|t) * P(X(j,t)|t) \!=\! w_{i,j} * rw^n \end{split}$$

where w_{ij} is the original weight of the interaction and n is the number (0–2) of lowly-expressed genes in tissue t out of $\{P_{ij}P_i\}$.

Thus, conversion of the generic PPI weight to a tissue specific PPI weight using the edge reweight method involves multiplying an edge's weight by nv if one of its adjacent genes is not expressed in the tissue, and by rw^2 if neither of the edge's adjacent genes are expressed in the tissue.

 $0 \leq rw \leq 1$.

Prioritization and performance evaluation

We extracted from GeneCards [39] 1347 gene-disease associations. 938 of these associations included diseases for which tissue association information was available. This narrowed gene-disease association set spans 744 distinct genetic disorders and 637 distinct causal genes. For most experiments, diseases with a maximal tissue association score (MAS) below a certain threshold were filtered out. Disease information was taken from the Online Mendelian Inheritance in Man (OMIM) knowledgebase [60]. The disease similarity network was constructed and pre-processed as described in [4]. PPI edge weights were also normalized by the degree of their adjacent protein. The algorithm parameters were the same as in [4]: $\alpha = 0.9$, c = -15 and 10 iterations.

To evaluate the performance of the different variants of PRINCE, we used a leave-one-out cross validation procedure. In each cross-validation trial, a single disease-gene association, $\langle g, d \rangle$ was removed from the association set. In addition, any other disease-gene association involving g was removed to avoid the trivial case where mutations in the same gene cause two very similar disorders. PRINCE was then executed to score the nodes of the network. For the purpose of performance assessment, we constructed an artificial genomic interval of 100 genes which are part of the generic network and are located around g on the genomic sequence, for every g. The scores assigned to these 100 genes were compared to g's score. Note that for NR networks, unexpressed genes may still appear in the artificial interval, but they automatically gain a score of '0'. Using an artificial linkage interval enabled us to simulate the real-life scenario where prioritization is done only on genes residing within the genomic interval association with a disease.

To generate the ROC curve, we bundled together all of the scores from all of the cross validation trials, sorting them from highest to lowest and recording true- and false- positive rates at various score cutoffs. The actual causal genes were considered positive, and the rest of the genes were considered negative.

For case-to-case rank comparison, we considered each trial separately, and counted in how many trials did the tissue-specific PRINCE gave the actual causal gene a better rank compared to the entire network PRINCE, in how many times tissue-specific PRINCE gave a worse rank, and in how many cases both input networks yielded the same rank (Table 1).

To assess the significance of the difference between the different AUCs, we employed 25-fold cross validation. We performed random partitions and used the standard deviation as error bars in Figure 3a. The statistical significance of the ranking differences was evaluated via Wilcoxon signed-rank tests (Tables 1 and S2).

To fine-tune the rw parameter, we constructed sets of TS-ERW networks for varying values of rw and repeated the leaveone-out cross validation procedure for each set, for both the entire disease-gene association set and the expressed disease genes association set. We filtered the diseases-gene association sets with a MAS>40% disease-tissue association threshold. We sampled the value of nv at constant intervals in the range [0,1]. Having observed that the AUC peaks at the lowest non-zero value of nv (0.1), we proceeded to sample smaller values of nv, each one smaller than the previous in an order of magnitude. We stopped this procedure when we observed a decline in performance at both sets and convergence to the AUC yielded by nv = 0.

Inferring disease-tissue associations

We filtered the disease-gene association set with a MAS>40% disease-tissue association threshold. The 40% threshold was picked in order to retain only high-confidence associations (\sim 90% estimated accuracy). We considered only disease-tissue associations where the causal gene is known to be expressed in the tissue assigned by Lage et al. [40].

For each disease-gene pair, we removed the association and ran PRINCE with the same definitions and parameters as the previous section. We repeated the procedure once per tissue, using that tissue's TS-ERW PPI with nv = 0.1 (A value shown to produce stable positive results for both association sets) as an input for PRINCE. We then assessed PRINCE's performance for every tissue using the **relative rank** PRINCE assigns the causal gene. Finally, we sorted the tissues' according to the PRINCE rank and ranked them accordingly.

We evaluated the correlation of our tissue ranking with the tissues given the highest association score by Lage et al. for each disease (denoted 'assigned tissue' from now on). For every disease-gene association, we checked the ranking we gave to the assigned tissue.

To provide an estimated p-value for the high number of highlyranked assigned tissues, we performed a permutation test as follows: For every disease-gene association, we assigned at random a tissue to the disease, selecting from the tissues where the causal gene is expressed (to counter the bias caused from focusing on disease-gene associations where the gene is expressed in the assigned tissue), and marked the ranking we give the randomly assigned tissue'. When using the 'ranking by PRINCE rank' scheme, we counted how many times the random tissue was ranked first. We repeated this procedure 1000 times.

Supporting Information

Dataset S1 Entrez ids of network genes. At Datasets S2, S3, the network genes are indexed from 1 to 9998. This file contains the Entrez ids of these genes, sorted by their index. (XLSX)

Dataset S2 The Edge-Reweight tissue-specific PPI network. The file is divided to 60 sections. Section headers are denoted with '#'. Each row represents a single interaction. The first and second columns are the interacting genes' indices, and the third column is the interaction confidence, after reweight. Confidences are not normalized. This network was generated with nv = 0.1.

(ZIP)

Dataset S3 The Node-Removal tissue-specific PPI network. The file is divided to 60 sections. Section headers are denoted with '#'. Each row represents a single interaction. The first and second columns are the interacting genes' indices, and the third column is the interaction confidence. Confidences are not normalized. Note that even though some of the genes are removed at each network, the gene indices are the same as in the other Datasets.

(ZIP)

Figure S1 Benchmarking the rw parameter. Comparing the ROC AUC obtained by a leave-one-out cross validation trials for varying values of nw, using (A) The expressed disease-genes association set and (B) the entire disease-gene association set. Disease-tissue associations were filtered using a MAS>40% threshold.

(PNG)

Figure S2 ROC curve comparison of generic and tissuespecific variants of PRINCE. These ROC curves yielded the ROC AUC values presented in Figure 3. The curves are the output of a leave-one-out cross validation test, using the expressed disease-genes association set and filtering disease-tissue associations with a MAS threshold of 40%. (PNG)

Figure S3 Comparing generic and tissue-specific PPIs' performance in disease genes prioritization using the entire disease-gene association set. Performance comparison between generic and different variants of tissue-specific PRINCE according to ROC Area Under Curve of causal gene prediction in a leave-one-out cross validation test, using the entire disease-gene association data set. The comparison also includes a special variant of generic PRINCE where genes unexpressed at the tissue get an automatic score of 0 (Orange column, described at the third paragraph of the discussion section). Test cases where disease-tissue association had a MAS lower than 40% were discarded.

(PNG)

Figure S4 Evaluation of tissue-disease association inference using the Absolute Score scheme. The histogram shows the distribution of our disease-tissue ranking for the tissues assigned by Lage et al, when we use the Absolute Score ranking scheme instead of the Relative Rank ranking scheme. In this scheme, tissues are ordered according to the score PRINCE assigns to the actual causal gene at every tissue. As can be seen, this scheme leads to a more fine-grained differentiation of tissue ranking.

 (\mathbf{PNG})

Figure S5 Comparing generic and tissue-specific PPIs' performance using post-process attenuation of unexpressed genes. A performance comparison between the generic and different variants of tissue-specific PRINCE, using a special version of PRINCE where, in a post-processing step, the scores of all genes not expressed in the relevant tissue is set to 0. These AUC values were obtained by a leave-one-out cross validation trial using the expressed disease-genes set and a MAS threshold of 40%. (PNG)

Table S1 Topological properties of the tissue-specific Node Removal networks. OULOW

(XLSX)

Table S2Evaluation of generic and tissue-specific geneprioritization methods using the entire disease-geneassociation set. The table presents a case-to-case comparison ofthe ranking provided by generic and tissue-specific PRINCE, aswell as the statistical significance of this comparison usingWilcoxon signed-rank test.(PDF)

Text S1 Supporting results. Describing the analysis of causal genes prioritizations with tissue-specific networks using the entire disease-gene association set, as well as disease-tissue association inference using the Absolute Score tissue ranking scheme. (PDF)

Acknowledgments

We would like to thank Kasper Lage for providing us the disease-tissue association matrix used in [40]. We would also like to thank Nir Yosef, Shiri Freilich, Liat Perlman and Nir Atias for their advice.

References

- Schunkert H, König IR, Kathiresan S, Reilly MP, Assimes TL, et al. (2011) Large-scale association analysis identifies 13 new susceptibility loci for coronary artery disease. Nat Genet 43: 333–338.
- Birnbaum S, Ludwig KU, Reutter H, Herms S, Steffens M, et al. (2009) Key susceptibility locus for nonsyndromic cleft lip with or without cleft palate on chromosome 8q24. Nat Genet 41: 473–477.
- Kathiresan S, Voight BF, Purcell S, Musunuru K, Ardissino D, et al. (2009) Genome-wide association of early-onset myocardial infarction with single nucleotide polymorphisms and copy number variants. Nat Genet 41: 334–341.
- Vanunu O, Magger O, Ruppin E, Shlomi T, Sharan R (2010) Associating genes and protein complexes with disease via network propagation. PLoS Comput Biol 6: e1000641.
- Köhler S, Bauer S, Horn D, Robinson PN (2008) Walking the interactome for prioritization of candidate disease genes. Am J Hum Genet 82: 949–958.
- Wu X, Jiang R, Zhang MQ, Li S (2008) Network-based global inference of human disease genes. Mol Syst Biol 4: 189.
- Oti M, Snel B, Huynen MA, Brunner HG (2006) Predicting disease genes using protein-protein interactions. J Med Genet 43: 691–698.
- Linghu B, Snitkin ES, Hu Z, Xia Y, Delisi C (2009) Genome-wide prioritization of disease genes and identification of disease-disease associations from an integrated human functional linkage network. Genome Biol 10: R91.
- Lage K, Karlberg EO, Størling ZM, Olason PI, Pedersen AG, et al. (2007) A human phenome-interactome network of protein complexes implicated in genetic disorders. Nat Biotechnol 25: 309–316.
- Karni S, Soreq H, Sharan R (2009) A network-based method for predicting disease-causing genes. J Comput Biol 16: 181–189.
- Gaulton KJ, Mohlke KL, Vision TJ (2007) A computational system to select candidate genes for complex human traits. Bioinformatics 23: 1132–1140.
- Franke L, van Bakel H, Fokkens L, de Jong ED, Egmont-Petersen M, et al. (2006) Reconstruction of a functional human gene network, with an application for prioritizing positional candidate genes. Am J Hum Genet 78: 1011–1025.
- Perez-Iratxeta C, Bork P, Andrade-Navarro MA (2007) Update of the G2D tool for prioritization of gene candidates to inherited diseases. Nucleic Acids Res 35: W212–216.
- George RA, Liu JY, Feng LL, Bryson-Richardson RJ, Fatkin D, et al. (2006) Analysis of protein sequence and interaction data for candidate disease gene prediction. Nucleic Acids Res 34: e130.
- Tiffin N, Andrade-Navarro MA, Perez-Iratxeta C (2009) Linking genes to diseases: it's all in the data. Genome Med 1: 77.
- Oti M, Brunner HG (2007) The modular nature of genetic diseases. Clin Genet 71: 1–11.
- Navlakha S, Schatz MC, Kingsford C (2009) Revealing biological modules via graph summarization. J Comput Biol 16: 253–264.
- Lee I, Blom UM, Wang PI, Shim JE, Marcotte EM (2011) Prioritizing candidate disease genes by network-based boosting of genome-wide association data. Genome Res 21: 1109–1121.
- Ideker T, Sharan R (2008) Protein networks in disease. Genome Res 18: 644– 652.
- Goh KI, Cusick ME, Valle D, Childs B, Vidal M, et al. (2007) The human disease network. Proc Natl Acad Sci U S A 104: 8685–8690.
- Winter EE, Goodstadt L, Ponting CP (2004) Elevated rates of protein secretion, evolution, and disease among tissue-specific genes. Genome Res 14: 54–61.
- Chao EC, Lipkin SM (2006) Molecular models for the tissue specificity of DNA mismatch repair-deficient carcinogenesis. Nucleic Acids Res 34: 840–852.
- Zhong Q, Simonis N, Li QR, Charloteaux B, Heuze F, et al. (2009) Edgetic perturbation models of human inherited disorders. Mol Syst Biol 5: 321.
- 24. Bossi A, Lehner B (2009) Tissue specificity and the human protein interaction network. Mol Syst Biol 5: 260.
- Emig D, Albrecht M (2011) Tissue-specific proteins and functional implications. J Proteome Res 10: 1893–1903.
- Lin WH, Liu WC, Hwang MJ (2009) Topological and organizational properties of the products of house-keeping and tissue-specific genes in protein-protein interaction networks. BMC Syst Biol 3: 32.
- Waldman YY, Tuller T, Shlomi T, Sharan R, Ruppin E (2010) Translation efficiency in humans: tissue specificity, global optimization and differences between developmental stages. Nucleic Acids Res 38: 2964–2974.
- Lopes TJ, Schaefer M, Shoemaker J, Matsuoka Y, Fontaine JF, et al. (2011) Tissue-specific subnetworks and characteristics of publicly available human protein interaction databases. Bioinformatics 27: 2414–2421.
- Jiang BB, Wang JG, Xiao JF, Wang Y (2009) Gene Prioritization for Type 2 Diabetes in Tissue-specific Protein Interaction Networks The Third International Symposium on Optimization and Systems Biology. Zhangjiajie, China: ORSC & APORC. pp. 319–328.

Author Contributions

Conceived and designed the experiments: OM YYW ER RS. Performed the experiments: OM YYW. Analyzed the data: OM. Contributed reagents/materials/analysis tools: OM YYW. Wrote the paper: OM YYW ER RS.

- Odom DT, Dowell RD, Jacobsen ES, Nekludova L, Rolfe PA, et al. (2006) Core transcriptional regulatory circuitry in human hepatocytes. Mol Syst Biol 2: 2006.0017.
- Amit I, Garber M, Chevrier N, Leite AP, Donner Y, et al. (2009) Unbiased reconstruction of a mammalian transcriptional network mediating pathogen responses. Science 326: 257–263.
- Alexopoulos LG, Saez-Rodriguez J, Cosgrove BD, Lauffenburger DA, Sorger PK (2010) Networks inferred from biochemical data reveal profound differences in toll-like receptor and inflammatory signaling between normal and transformed hepatocytes. Mol Cell Proteomics 9: 1849–1865.
- Chikina MD, Huttenhower C, Murphy CT, Troyanskaya OG (2009) Global prediction of tissue-specific gene expression and context-dependent gene networks in Caenorhabditis elegans. PLoS Comput Biol 5: e1000417.
- Hu J, Wan J, Hackler L, Zack DJ, Qian J (2010) Computational analysis of tissue-specific gene networks: application to murine retinal functional studies. Bioinformatics 26: 2289–2297.
- Shlomi T, Cabili MN, Herrgård MJ, Palsson B, Ruppin E (2008) Network-based prediction of human tissue-specific metabolism. Nat Biotechnol 26: 1003–1010.
- Jerby L, Shlomi T, Ruppin E (2010) Computational reconstruction of tissuespecific metabolic models: application to human liver metabolism. Mol Syst Biol 6: 401.
- Gille C, Bölling C, Hoppe A, Bulik S, Hoffmann S, et al. (2010) HepatoNet1: a comprehensive metabolic reconstruction of the human hepatocyte for the analysis of liver physiology. Mol Syst Biol 6: 411.
- Su AI, Wiltshire T, Batalov S, Lapp H, Ching KA, et al. (2004) A gene atlas of the mouse and human protein-encoding transcriptomes. Proc Natl Acad Sci U S A 101: 6062–6067.
- Rebhan M, Chalifa-Caspi V, Prilusky J, Lancet D (1997) GeneCards: integrating information about genes, proteins and diseases. Trends Genet 13: 163.
- Lage K, Hansen NT, Karlberg EO, Eklund AC, Roque FS, et al. (2008) A largescale analysis of tissue-specific pathology and gene expression of human disease genes and complexes. Proc Natl Acad Sci U S A 105: 20870–20875.
- Yook SH, Oltvai ZN, Barabási AL (2004) Functional and topological characterization of protein interaction networks. Proteomics 4: 928–942.
- Pontén F, Gry M, Fagerberg L, Lundberg E, Asplund A, et al. (2009) A global view of protein expression in human cells, tissues, and organs. Mol Syst Biol 5: 337.
- Rocke DM, Durbin B (2001) A model for measurement error for gene expression arrays. J Comput Biol 8: 557–569.
- Quackenbush J (2002) Microarray data normalization and transformation. Nat Genet 32 Suppl: 496–501.
- Blake WJ, KAErn M, Cantor CR, Collins JJ (2003) Noise in eukaryotic gene expression. Nature 422: 633–637.
- Arteaga MF, Coric T, Straub C, Canessa CM (2008) A brain-specific SGK1 splice isoform regulates expression of ASIC1 in neurons. Proc Natl Acad Sci U S A 105: 4459–4464.
- 47. Vogel C, Abreu ReS, Ko D, Le SY, Shapiro BA, et al. (2010) Sequence signatures and mRNA concentration can explain two-thirds of protein abundance variation in a human cell line. Mol Syst Biol 6: 400.
- Takeyama K, Kitanaka S, Sato T, Kobori M, Yanagisawa J, et al. (1997) 25-Hydroxyvitamin D3 lalpha-hydroxylase and vitamin D synthesis. Science 277: 1827–1830.
- Sharan R, Ulitsky I, Shamir R (2007) Network-based prediction of protein function. Mol Syst Biol 3: 88.
- Vazquez A, Flammini A, Maritan A, Vespignani A (2003) Global protein function prediction from protein-protein interaction networks. Nat Biotechnol 21: 697–700.
- Su AI, Cooke MP, Ching KA, Hakak Y, Walker JR, et al. (2002) Large-scale analysis of the human and mouse transcriptomes. Proc Natl Acad Sci U S A 99: 4465–4470.
- Ochiai A (1957) Zoogeographic studies on the soleoid fishes found in Japan and its neighbouring regions. Bull Jpn Soc Sci Fish 22: 526–530.
- Huang dW, Sherman BT, Lempicki RA (2009) Systematic and integrative analysis of large gene lists using DAVID bioinformatics resources. Nat Protoc 4: 44–57.
- Dennis G, Sherman BT, Hosack DA, Yang J, Gao W, et al. (2003) DAVID: Database for Annotation, Visualization, and Integrated Discovery. Genome Biol 4: P3.
- Rual JF, Venkatesan K, Hao T, Hirozane-Kishikawa T, Dricot A, et al. (2005) Towards a proteome-scale map of the human protein-protein interaction network. Nature 437: 1173–1178.
- Stelzl U, Worm U, Lalowski M, Haenig C, Brembeck FH, et al. (2005) A human protein-protein interaction network: a resource for annotating the proteome. Cell 122: 957–968.

- Ewing RM, Chu P, Elisma F, Li H, Taylor P, et al. (2007) Large-scale mapping of human protein-protein interactions by mass spectrometry. Mol Syst Biol 3: 89.
 Peri S, Navarro JD, Kristiansen TZ, Amanchy R, Surendranath V, et al. (2004)
- Peri S, Navarro JD, Kristiansen TZ, Amanchy R, Surendranath V, et al. (2004) Human protein reference database as a discovery resource for proteomics. Nucleic Acids Res 32: D497–501.
- Bader JS, Chaudhuri A, Rothberg JM, Chant J (2004) Gaining confidence in high-throughput protein interaction networks. Nat Biotechnol 22: 78–85.
- Hamosh A, Scott AF, Amberger J, Bocchini C, Valle D, et al. (2002) Online Mendelian Inheritance in Man (OMIM), a knowledgebase of human genes and genetic disorders. Nucleic Acids Res 30: 52–55.